Contents lists available at ScienceDirect





journal homepage: www.elsevier.com/locate/crm

A stepwise approach for identifying climate change induced socio-economic tipping points

Kees C.H. van Ginkel^{a,b,*}, Marjolijn Haasnoot^{a,c}, W.J. Wouter Botzen^{b,d,e}

^a Deltares, Delft, The Netherlands

^b Institute for Environmental Studies (IVM), VU University, Amsterdam, The Netherlands

^c Utrecht University, Utrecht, The Netherlands

^d Utrecht School of Economics (U.S.E.), Utrecht University, Utrecht, The Netherlands

^e Risk Management and Decision Processes Centre, The Wharton School, University of Pennsylvania, USA

ARTICLE INFO

Keywords: Tipping points Regime shifts Transformational adaptation Decision making under deep uncertainty Exploratory modelling Sea level rise Coastal city

ABSTRACT

Climate change may cause socio-economic tipping points (SETPs), where the state of a socioeconomic system abruptly changes to a fundamentally different state. While their potential existence is recognized, a systematic method for policy-relevant research on SETPs is lacking. This study introduces a stepwise approach for identification of SETPs that supports decision making under uncertain climate and socio-economic conditions. The approach is demonstrated with a stylized case study on the collapse of house prices (a SETP) in a coastal city, due to increasing flood risk from sea level rise. We explore four dynamic adaptive management strategies under a wide range of possible futures. We find that under scenarios with very high and rapid sea level rise, tipping points in real estate prices occur if the market responds to sudden changes in perceived flood risk rather than gradually adjusting prices to changes in flood risk in a rational manner. Such SETPs can only be avoided with a proactive strategy and when flood protection measures are implemented rapidly. Our approach can guide future studies on SETPs and seeks to move the study of SETPs towards a concept that provides perspective of action for policy makers.

1. Introduction

Climate change can have large impact on society and economy (IPCC, 2014). Gradual climate change can cause societal disruptions and large economic damage (Carleton and Hsiang, 2016; Nordhaus, 2018). In addition, climate tipping points (Box 1) can have dire socio-economic consequences (Lemoine and Traeger, 2016; Lontzek et al., 2015; Dietz et al., 2021). Much less attention has been paid to cases where the climate changes gradually, but the socio-economic system responds in a non-linear and abrupt fashion (Biggs et al., 2018; Kopp et al., 2016). Such climate change induced socio-economic tipping points (SETPs) are points where the state of a socio-economic system abruptly changes to a fundamentally different state, following relatively marginal climate change (van Ginkel et al., 2020). It is important for adaptation and mitigation policy development to understand if and how SETPs may occur.

Following Van Ginkel et al. (2020), this study defines a socio-economic tipping point (SETP) as a point of abrupt change of a socioeconomic system into a new, fundamentally different state, using three criteria, see box 1. Two SETP subtypes are *impact* and *transformative response* tipping points (Maier et al., 2021; Winkelmann et al., 2022; van Ginkel et al., 2020), see box 1. An example of an

* Corresponding author at: Deltares, Boussinesqweg 1, 2629 HV Delft, The Netherlands. *E-mail address*: Kees.vanGinkel@Deltares.nl (K.C.H. van Ginkel).

https://doi.org/10.1016/j.crm.2022.100445

Received 1 December 2021; Received in revised form 30 June 2022; Accepted 8 July 2022 Available online 12 July 2022





^{2212-0963/© 2022} The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

K.C.H. van Ginkel et al.

impact-SETP is that at some degree of sea level rise, a low-altitude atoll island may suddenly become uninhabitable due to diminishing fresh water resources, increasing infrastructure damage during storm surge events, unaffordability of flood protection infrastructure and eventually permanent inundation (Magnan and Duvat, 2020; Storlazzi et al., 2018; Barnard et al., 2021). A *transformative response* SETP would be the proactive retreat from the islands before these impacts happen (Bordner et al., 2020, Siders et al., 2019). It is important for spatial-economic planners and capital investors to know if and under what conditions SETPs may happen, and what can be done to anticipate their causes and effects. However, the occurrence of SETPs is very uncertain due to complex systems interactions, model uncertainty, and uncertainty on future climate, socio-economic developments and adaptation (Biggs et al., 2018; van der Bolt et al., 2018; Kopp et al., 2019; Marchau et al., 2019).

In the emerging literature on tipping points in coupled socio-ecological systems most tipping points occur in the biophysical rather than the socio-economic realm (Filatova et al., 2016; Milkoreit et al., 2018). The few model studies describing climate change induced tipping points in the socio-economic domain (e.g. de Koning and Filatova, 2020; Hossain et al., 2017; Kabir et al., 2018) typically investigate a small set of possible futures or ignore adaptive policy change, which are highly relevant on longer time-scales when climate and socio-economic scenarios increasingly diverge, and adaptation to these new conditions can be expected (Biggs et al., 2018). Moreover, a perspective of action for decision makers is often missing (Verburg et al., 2016). What is lacking is a systematic stepwise approach for identification of climate change-induced SETPs, which accounts for uncertainties, many possible futures, and adaptive policy change and which can be used to support climate adaptation and mitigation policy.

Building blocks for such an approach can be found in the *Decision Making under Deep Uncertainty (DMDU)* research field dedicated to decision support in conditions of uncertain and complex system interactions (Marchau et al., 2019). *Decision Scaling* (Brown et al., 2012) and *Robust Decision Making* (Groves and Lempert, 2007) use computational techniques like *Scenario Discovery* to identify clusters of outcomes of interest under many possible futures (Bryant and Lempert, 2010; Kwakkel, 2017). Outcomes of interest may be threshold values in the model outcomes, time series with distinct model behaviour (Steinmann et al., 2020), or as we will show in this paper: SETPs. Identification of SETPs does not only require analysing when an indicator exceeds threshold conditions, but also requires identifying an abrupt shift from one to another stable state (e.g. Andersen et al., 2009; Truong et al., 2020).

The objective of this paper is to present a stepwise approach that identifies under what conditions SETPs may occur and guides policy aimed at preventing adverse SETPs or supporting positive SETPs. This is done by bringing together DMDU and tipping point literature in a stepwise approach and illustrating it with a stylized case study on flood risk in a coastal city like Rotterdam. We develop a model with characteristics of both a system-dynamic and an agent-based model, that simulates flood risk and house prices under

Box 1

Conceptualisation of climate change induced socio-economic tipping points

In contrast to their biophysical counterparts, where the tipping element is a (sub)system of the climate (Lenton et al., 2008) or ecosystem (Scheffer and Carpenter, 2003), the tipping element here is a delineated socio-economic system, and climate change is a driver (Fig. 1). After Milkoreit et al. (2018) and Van Ginkel et al. (2020), we define a socio-economic tipping point (SETP) as a point of: rapid, abrupt change compared to typical rates of change in the system (Criterion 1); indicating a transition from one to another relatively stable state (Criterion 2); these states being fundamentally different (Criterion 3). Strict system-dynamic conceptualisations such as mathematical bifurcations satisfy these criteria, while they also leave room for a more qualitative conceptualisation where argumentation shows that these three criteria are satisfied.

Two types of SETPs (Van Ginkel et al., 2020; Maier et al., 2020) are impact and response tipping points (Fig. 1). Impact SETPs refer to cases where the state shift occurs due to autonomous, endogenous system dynamics and insufficient human action. Some examples are: slow-onset climate hazards tipping a region to a reinforcing system state where people seek to migrate out of the region (Kopp et al., 2016; Kabir et al., 2018; McLeman et al., 2017); increasing flood risk causing a collapse of an established natural hazard insurance system (Tesselaar et al., 2020); or a retreating snowline causing bankruptcy of low-altitude ski resorts (Vaghefi et al., 2021). These examples are predominantly 'negative', given their close connection to climate change impact and natural disaster impact literature, but in principle they can also be 'positive', e.g. when the new climate conditions open new opportunities for agriculture. One reason for SETPs may be that critical limits to human adaptation are reached (Thomas et al., 2021).

Response SETPs refer to cases where the socio-economic structure is deliberately transformed by human action, in anticipation to foreseen impacts. Rooted in innovation and change theory (e.g. Rogers, 1962), and popularized by Gladwell (2000), this literature is predominantly 'positive' and seeks to benefit from non-linear system change to achieve abrupt, fundamental societal change to meet adaptation and mitigation challenges (Burch et al., 2017, Tabara et al., 2021). Transformative adaptation refers to adaptation actions that fundamentally change a socio-economic system (Fedele et al., 2019; Garschagen and Solecki, 2017; Kates et al., 2012; Berrang-Ford et al., 2021). Transformative mitigation, often referred to as 'social tipping', refers to rapid, contagious spreading of behaviours, opinions, knowledge, technology, and social norms causing a shift to a socio-economic system with strongly reduced carbon emissions (Otto et al., 2020; Stadelmann-Steffen et al., 2021; Winkelmann et al., 2022).

Regardless of its negative connotation in climate change impact literature, we explicitly intend to include positive system shifts as well (Otto et al., 2020; Tàbara et al, 2018). In fact, the desirability of socio-economic system change is case-specific and subjective: the same tipping point might be opportune for some stakeholders that benefit from the new state, but at the same time be destructive for others that depend on the old state (Young, 2012).



Fig. 1. Types of tipping points positioned in the archetypical cause-effect chain from greenhouse gas emissions to the climate and other natural systems, to socio-economic system impacts and human response. Source: Van Ginkel et al., 2020, CC BY License.



Fig. 2. Overview of the stepwise approach. The XLRM-framework (Kwakkel, 2017; Lempert et al., 2005) distinguishes between eXternalities (exogenous drivers), policy Levers (possible system interventions), system Relations (model equations) and Metrics (indicators for system performance), see Fig. 4. SETPs refer to climate-change induced socio-economic tipping points. Grey arrows indicate optional iterations over steps.

uncertain SLR, storm surges, house market behaviour and dynamic adaptive policy responses. Section 2 outlines the stepwise approach. Section 3 introduces the stylized case. Section 4 applies the stepwise approach to the case study. Section 5 discusses how the approach can be applied to other global cases. Section 6 concludes.

2. Stepwise approach

The stepwise approach to identification and decision support on SETPs is outlined in Fig. 2, which is described in more detail below and summarized here. Step 1–3 delineate and describe the socio-economic system and its possible SETPs. Step 4 identifies the conditions under which SETPs occur. Step 5–7 further assess the impacts of SETPs and formulate a policy advice for decision makers on adaptation and mitigation.

2.1. Step 1: System delineation

The first step delineates the system boundaries, involving a demarcation of the spatial and temporal borders, but also of the system aspects to be considered. These may include physical, ecologic, social, economic, institutional, political or ethical aspects. System delineation (as well as system description and tipping point definition) inevitably reflects a certain actor perspective; an understanding of how the world works (Karstens et al., 2007). Transparent description of this perspective is important because different actors might have different perspectives on how the system works and what are important system indicators.

Typically, the clearest SETP examples can be found on small scales in homogeneous systems, because of their uniform response to drivers of change (Scheffer et al., 2012) and limited possibilities for economic substitution (Van Ginkel et al., 2020).

2.2. Step 2: System description

The second step describes the system with the XLRM framework (applied on the case study in Fig. 4). This framework guides long-



Fig. 3. Stylized presentation of the case study. Each year, the city experiences an extreme water level: the combined effect of sea level rise and a storm surge height. Residential area A is not protected, residential area B is protected by a dike which can be heightened due to adaptation. In both areas, the (perceived) flood risk is discounted in the house price.

term policy analysis by distinguishing (Lempert et al., 2005):

- Exogeneous uncertainties (X): the factors beyond control of decision makers on the system, such as climate change (RCPs) and external socio-economic development (SSPs);
- Policy levers (L): the adaptation policy actions or strategies for management of the system;
- Relations (R): the model linking exogeneous uncertainties and policy levers to outcomes of interest;
- and Metrics (M): indicators of system performance of interest to the decision maker, such as socio-economic or risk indicators.

Like the exogenous uncertainties (X), the policy levers (L) are external to the system, because they reflect the alternative strategies the decision maker can impose on the system from the outside. They may range from small incremental alternatives to the baseline strategy, to large transformative changes. Some measures may be so transformative, that their mere implementation could qualify as an SETP in terms of *transformative response* (Tabara et al., 2021, also see box 1). However, such response SETPs are likely to be a priori assumed, and not a posteriori identified in step 4, unless they result from surprising model outcomes (e.g. resulting from complex behaviour in agent-based models), cf. Maier et al. 2020.

One complication in the study of tipping points in socio-economic economic systems (in contrast to their biophysical counterparts) is that human actors consciously and continuously adapt the system to change (Biggs et al., 2018; Van Ginkel et al., 2020). This raises the question whether adaptation should be understood as external or endogenous; as having a cause outside or within the system. The context for which the approach is developed, is the situation where a decision (or policy) maker is confronted with a potential SETP in a socio-economic system they are responsible for. The policy alternatives at the disposal of this decision maker may be modelled as completely independent of the system dynamics; in this case they are described as external policy levers (L). However, they can also be described as part of the system itself; in that case they are part of the model relations (R).

The model relations (R) can be described using statistical, system-dynamic, graph-theoretic, agent-based or computational (partial/general) equilibrium models (Filatova et al., 2016), or any hybrid combinations. Note that the non-linear tipping behaviour often results from reinforcing and stabilizing system feedback loops (Milkoreit et al., 2018).

2.3. Step 3: Tipping point operationalisation

The third step operationalises the SETP in terms of performance metrics (M) of the system, by showing how tipping point occurrence can be identified from system metrics, using the following SETP-characteristics (Milkoreit et al., 2018; Van Ginkel et al., 2020):

- C1. rapid, abrupt change compared to typical rates of change in the system;
- C2. indicating a transition from one relatively stable state before (B) to another relatively stable state after (A);
- C3. with B being fundamentally different from A.

The operationalisation captures the non-linear, S-curve behaviour of the indicator metric timeseries. One method is to detect early



Fig. 4. Experimental set-up. Three exogenous uncertainties (X) and one policy lever (L) feed into the coupled physical-socioeconomic model (R), which has two alternative formulations for dynamics on the housing market. Each unique combination of X, L and R results in two timeseries with house prices for both residential areas (M). These timeseries are analysed to identify SETPs, described in the SETP-metric M*. See SI for model details.

warning signals of an approaching tipping point in the timeseries such as increasing autocorrelation, increasing variance, increasing skewness and flickering between two stable states (Scheffer et al., 2009; Singh et al., 2018). Another method is to detect the tipping point itself with a statistical test. For example, the null-hypothesis is a model with constant mean and variance for the entire metric time series, and the alternative hypothesis is a model with two states with different mean and variance which are separated by a step/ change point (Carstensen and Weydmann, 2012). These change points can be detected with various algorithms (e.g. Andersen et al., 2009; Quandt, 1958, Truong et al., 2020). Yet another method is to tailor the three tipping point criteria (box 1) specifically to the metric time series, as done in our case study (Section 4.3).

2.4. Step 4: Tipping point identification

The fourth step identifies what model inputs cause tipping points in model outputs in the many possible futures, through an exploratory analysis (Moallemi et al., 2020). For example, scenario discovery techniques can identify for which combinations of input parameters (uncertainties X and policy levers L) the model behaviour of interest occurs (Hadjimichael et al., 2020; Kwakkel, 2017; Moallemi et al., 2020). Fig. 4 shows that to use these techniques, one intermediate step needs to be taken. The outcome timeseries (M (t)) of each experiment (an experiment is a model run with a unique combination of X, L and R) is tested on the occurrence of SETPs with the algorithm defined in the previous step. The results of this step are saved as a new metric M*.

2.5. Step 5: Impact assessment

The fifth step assesses the socio-economic system impacts of the identified tipping points. While step four only identified under which external circumstances (X and L) SETPs do occur, step five assesses how tipping has changed the socio-economic system. This starts within the initial system boundaries, by evaluating the state of model variables after the tipping point. For example, when the model relations (R) are described with an agent-based model, one can examine what the socio-economic conditions after the tipping point imply for the different actors represented in the model. Or, when using a system-dynamic model, one could evaluate the model variables after the tipping point, and reflect on what they tell about the new state of society and the economy.

Additionally, impacts beyond the initial system boundaries can be explored, by widening the spatial boundaries, reasoning beyond the time horizon, or beyond the aspects included in the initial analysis. For example, if the tipping point involves the collapse of a company or industry, the impact on the wider economic sector can be studied with a sectoral-economic or macro-economic model. If one suspects that the local SETP may also cause a tipping point on a larger scale, the feedback loop from step five to step one could be triggered (Fig. 2), repeating the analysis with widened system boundaries.

2.6. Step 6: Action identification

The sixth step identifies the possible actions that a decision maker can take to anticipate or avoid SETPs, or to mitigate the impacts of their occurrence. This starts by assessing the effectiveness of the policy levers that were included in the model. When possible new actions are identified, the feedback loop from step six to two can optionally be triggered, in which the actions are added as new policy levers, and their effectiveness is evaluated. An action can also be to monitor the variables that provide an early warning of system change.

2.7. Step 7: Decision support

In the last step, the lessons learned are translated to adaptation, mitigation and monitoring policies. Arguably, adaptation is most relevant for local decision makers since it directly limits local climate risk, while mitigation depends on actions that are typically taken by other jurisdictions. When in a qualitative approach concrete policy levers were included, one can reflect on their effectiveness to prioritize actions. We further recommend reflecting on the conditions that led to SETPs. This can inform policy to monitor certain key system indicators providing early warning of SETPs, and to timely intervene in the system.

3. Introduction to case study

Sea level rise is a global threat to low-lying coastal cities (Oppenheimer et al., 2020). Flood damage and exposed population are projected to strongly increase (Brown et al., 2018; Neumann et al., 2015; Vitousek et al., 2017), even with large adaptation investments to maintain constant flood probability (Hallegatte et al., 2013). It is relevant to explore the many possible futures of coastal cities on the occurrence of SETPs because SLR-projections are uncertain due to possible ice-sheet instability and increased mass-loss from Antarctica (Bamber et al., 2019; Horton et al., 2020; Le Bars et al., 2017). Moreover, coastal cities may experience non-linear impacts and possibly tipping points from increasing flood risk (de Koning et al., 2019; de Koning and Filatova, 2020; Haasnoot et al., 2020; Taherkhani et al., 2020; Barnard et al., 2021). Rotterdam is an interesting case for illustrating our stepwise approach in the context of this global challenge, because most of the city is exposed to coastal flooding while also having a high protection level. Moreover, storylines leading to possible SETPs for the city have been proposed in scientific literature (Haasnoot et al., 2020; Olsthoorn et al., 2008; van Ginkel et al., 2020), by the Dutch central bank (Caloia and Jansen, 2021), and in public debate (Bregman, 2020), but as of yet a systematic quantitative assessment of SETPs is lacking.

Rotterdam is the second largest city in The Netherlands and, located in the Rhine-Meuse Delta, it hosts the largest port of Europe.

Dikes and storm surge barriers protect the polders against coastal and riverine floods, protecting against a 1:10,000 year flood. Besides the low-altitude polders (-2 m), Rotterdam also has significant portions of unprotected areas which are elevated just above mean sea level (+3 to 5 m). Earlier studies found no technical or financial barriers for Rotterdam's flood protection for a SLR up to 105 cm SLR by 2100, although around 40–50 cm, the Maeslant storm surge barrier might no longer suffice, which requires new actions (Kwadijk et al., 2010). Under more extreme SLR scenarios, the current strategy might need more radical transformation, and the implementation times of the associated measures might become a limiting factor (Haasnoot et al., 2020). Explorations of a 5 m SLR scenario confirmed that constraints to coastal adaptive capacity were not technical, engineering or cost-benefit in nature, but rather social and political (Olsthoorn et al., 2008; cf. Hinkel et al., 2018). Experts suggest possible decreasing investments, hampered economic development and



Fig. 5. Illustrative runs to demonstrate model behaviour. Panel 1 shows two extreme water level scenarios: each scenario is the sum of a storm surge and sea level rise scenario. Each extreme water level scenario is combined with two management strategies, the panel also shows the timing and resulting dike heights from dike heightening measures. Panels 2AB show the flood damage and near miss history of the small unembanked area A (Panel 2A, 21st century) and the embanked city centre area B (Panel 2B, 22nd century). Panels 3AB show the flood risk perception of residents of both areas. Panels 4AB indicate the development of the house prices when assuming a rational versus a boundedly rational housing market.

falling house prices in the areas exposed to large flood risk, potentially resulting in SETPs (Olsthoorn et al., 2008; van Ginkel et al., 2020).

4. Application of steps to case study

To apply the stepwise approach to the case study, we developed a stylized coupled physical and socio-economic model for an archetypical city with characteristics of Rotterdam, from which lessons can be learned for other cities worldwide. It simulates long-term development of flood risk and house prices under various dynamic adaptive strategies and many possible futures, as described in the SI (Section SI 1), also see 'Data and code availability'.

4.1. Step 1: System delineation

Spatially, the system is demarcated as two residential areas of Rotterdam (Fig. 3). Residential area A is elevated 3 m above mean sea level and is only protected by a small 0.5 m flood wall and is further considered as 'unembanked'. Residential area B is 1 m below mean sea level and is embanked with a 4.5 m high dike that can be heightened (Figure SI 1). Temporally, the future of the region is explored from 2020 till 2200 to capture the long-term impact of adaptation decisions and the effect of a possible acceleration of SLR in the 21st century (Le Bars et al., 2017). The focus is on these system aspects: development of flood risk, flood risk perceptions, and their impact on the house price, aggregated per residential area. We restrict ourselves to public sector adaptation measures, which ignores household adaptation like insurance or floodproofing. In the Netherlands, it is currently not possible to buy insurance against coastal flooding. In the polder situation (Residential area B), storm surge water levels may easily reach 3–5 m above polder ground level (see Fig. 5, panel 1). Therefore, household floodproofing measures are (nowadays) rarely applied, making flood protection completely dependent on dikes. For simplicity, we kept the flood wall at Residential Area A constant, to create a reference case without adaptation.

4.2. Step 2: System description

The core of our storyline is that flood risk affects the house price, as is supported by empirical research (Beltrán et al., 2018; Bin and Landry, 2013; Bosker et al., 2019). The flood risk increases due to SLR but can be reduced by dike heightening measures taken by the government. Flood and near-miss events abruptly elevate the risk perception of residents, which declines over the course of time if nothing happens (Bin and Landry, 2013; McNamara and Keeler, 2013; Tonn and Guikema, 2018).

Structuring the system along XLRM (Fig. 4), there are three external uncertainties (X). The magnitude and rate of sea level rise (X_1) depends on (amongst others) the RCP scenario and the potential collapse of the icesheets, see Figure SI 3 (Bamber et al., 2019; Horton et al., 2020; Le Bars et al., 2017). The height of storm surge events (X_2) are stochastically drawn from an extreme value distribution (Sterl et al., 2009), on a yearly basis. The implementation times of flood protection measures (X_3) can also vary (Haasnoot et al., 2020). The system can be managed by one policy lever (L_1) which can be set to four archetypical public sector flood risk management strategies, reflecting a continuum from reactive to proactive hard protection strategies (Oppenheimer et al., 2019). The 'reactive' strategy acts upon near miss or flood events. The 'economic' strategy anticipates the objective flood risk; the 'sentiment' strategy the perceived flood risk. The 'proactive' strategy seeks to maintain dike protection levels on ambitious target levels (Table SI 4).

The model relations (R) are illustrated with example model runs in Fig. 5 and summarized in Figure SI 2. The storm surge heights are combined with the sea level rise scenario, giving an extreme water level per yearly timestep (Fig. 5, panel 1), Section SI 1.1. Protection against these extremes in area A is given by the elevation of the residential area, and in area B by a dike, which can be heightened by adaptation. In case of a flood, the damage to the unembanked area (panel 2A) or the embanked area (panel 2B) is calculated. These panels also show any near-miss events in which floods nearly occurred, Section SI 1.2. Flood and near-miss events heighten the risk perception per area (panel 3A and 3B); these perceptions decline in years when nothing happens (Aerts, 2020), Section SI 1.4. The model relations have one variable component (R_1): the behaviour of the housing market. The rational housing market discounts the *actual flood risk* in the house price, whereas the boundedly rational housing market discounts the *perceived flood risk* in the house price (Fig. 5, panel 4A and 4B), Section SI 1.5. The perceived flood risk may strongly deviate from the actual flood risk due to the experience of flood events and social interactions between the areas (Botzen et al., 2009; Haer et al., 2017; Tonn et al.,

Table 1

Formalisation of tipping points criteria, tailored to the Dutch house market, see SI for details.

| Criterion | C1: Rapid, abrupt change | C2: Stable states | C3: Fundamentally different states |
|------------------------------|---|---|---|
| Model indicator threshold | The first-order derivative of the house price is above a threshold | The variance in the states before (B) and after (A) are above a threshold | The mean house prices in the state after and before are substantially different |
| Mathematical implementation | $M_{RA}^{\prime}(t) = rac{dM_{RA}(t)}{dt} ig M_{RA}^{\prime}(t) ig angle c_1$ | $\widehat{\sigma}_t^2ig(m{W}_{t,m{B}}ig) > c_2 \cap \widehat{\sigma}_t^2ig(m{W}_{t,m{A}}ig) > c_2$ With: | $\frac{ W_{t,B} - W_{t,A} }{W_{t,B}} > c_3$ With: |
| Selected thresholds | $c_1 = 0.15M(t=0)_{RA}$ | $ \begin{array}{l} W_{t,B} = \{ M_{t-d-n}, M_{t-d-n+1}, \cdots, M_{t-d} \} \\ W_{t,A} = \{ M_{t+d}, M_{t+d+1}, \cdots, M_{t+d+n} \} \\ c_2 = 10^9 \ell^2 \\ d = 3 year \\ n = 4 year \end{array} $ | $\overline{W_t} = \frac{1}{n} \sum W_t$ $c_3 = 0.1$ |

2019). The objective flood risk only depends on the state of the flood protection, whereas the perceived flood risk combines the objective flood risk with the risk perception, Section SI 1.4.

The model is evaluated using two metrics (M): the development of the house price in the outer-dike area A (M_A) and the house price in the embanked city centre (M_B). Additional illustrative model runs highlight characteristic behaviour under different management strategies (Figure SI 7,8), storm surges scenarios (Figure SI 9) and implementation times (Figure SI 10).

4.3. Step 3: Tipping point operationalisation

We operationalise a SETP as a timestep *t* with rapid, abrupt change (C1) separating two substantially different (C3), stable states (C2), in the house price $M_{RA,t}$, of residential area *RA*. These three criteria are tailored to the Dutch housing market (Table 1, Figure SI 14). Compared to observed house price changes (Section SI 3.1), a yearly change of more than 15 % is considered rapid, abrupt change (C1). A persistent change of more than 10 % is considered substantially different (C3), calculated over a 4-year rolling time window *W*. The variance, calculated over the same time window, is compared to a model-specific threshold that was identified iteratively. The tipping point detection algorithm is described in Section SI 3.1.

Fig. 6 shows that not all large house price fluctuations correspond to tipping points. Panel A shows a gradual declining house price, without rapid change in any of the years (C1 not met). Panel B shows a point of rapid change, with a stable state before, but not after the tipping point (C2 not met). Panel C shows two points of rapid change with stable states before and after, however, in both cases these states are not substantially different (C3 not met). Panel D shows three perfect examples of SETPs (C1-C3 met).

In the decision context of a coastal city, an abrupt shift to a period of large house price instability (Fig. 6B) might be just as policy relevant as price collapse to a stable state (Fig. 6D). Therefore, in the tipping point identification phase, we also include these rapid transitions from stable to unstable states (Fig. 6B).



Fig. 6. Four examples of how the SETP-identification algorithm tests if points in the metric house price time series M(t) meet the tipping point criteria (C1-C3). Panel A results from a gradual risk increase in a rational housing market under low sea level rise. This happens in unembanked area A where no adaptation takes place, and eventually so much damage is expected that houses are worth nothing. Panel B results from slow and reactive adaptation to fast sea level rise, in combination with a boundedly rational housing market. When a flood disaster strikes, a strongly elevated risk perception causes a house price crash in Area B. The disaster triggers adaptation, but this is not fast enough to keep up with sea level rise. The house price temporarily restores upon completion of new measures, and drops again when new floods or near misses hit the city. Panel C results from a boundedly rational housing market under very low sea level rise in the unembanked area A. As displayed by the increasingly deep downward spikes, there is a gradual increase in risk, that only has a temporarily effect on prices after a near miss event or flood. House prices recover because risk perception declines over time. Panel D results from no adaptation to a high sea level rise in area A, in combination with a boundedly rational housing market. Floods or near miss events cause abrupt house price drops to significantly lower levels. House prices do only slowly recover because the risk keeps increasing while trust restores. When new events strike, house prices drop to even lower levels. See Section SI 3 for more details.

4.4. Step 4: Tipping point identification

Each of the 396,000 experiments selects a unique combination of a sea level rise scenario X_1 , a storm surge series X_2 , a measure implementation time X_3 , flood protection strategy L_1 and a housing market model R_1 (Fig. 4). Each experiment delivers two time series, the house price in area A $M_A(t)$ and area B $M_B(t)$, in which tipping points are identified (Section 4.3). The EMA-workbench (Kwakkel, 2017) was used to systematically detect combinations of input-parameters resulting in time series with tipping points, and to create a dimensional stack visualization (Suzuki et al., 2015) (Fig. 7). Additionally, the SI presents the results of two alternative techniques for exploratory analysis: feature scoring (Section SI 4.1) and scenario discovery (Section SI 4.2).

In the most extreme SLR scenarios (SLR scenario 1 and 2, Fig. 7), SETPs in the embanked City Centre occur under all four management strategies. Combined with a reactive strategy, SETPs occur even with very short measure implementation times (4 years for a small measure). The rate of SLR in these scenarios is too high to adequately anticipate with a reactive flood risk strategy. With the most proactive strategy, however, SETPs can be avoided if measures can be quickly implemented, i.e. within 7 years or faster. The proactive strategy best resembles the current flood protection strategy in Rotterdam. Observed implementation times of small dike heightening measures are typically shorter than 10 years, whereas the construction of large storm surge barriers may take 20–40 years (Haasnoot et al., 2020).



Fig. 7. Dimensional stack visualization of combinations of input parameters leading to tipping points in the housing price in the City Centre. The yaxis captures the uncertainty in SLR scenario (X_1) and the housing market (R_1) which may be rational (R) or boundedly rational (BR). The x-axis captures the uncertainty in the adaptation strategy (L_1) and the measure implementation time (X_3). Each grid-cell indicates the percentage of storm surge scenarios (X_2) causing tipping points in the house-price before 2200. Coloured grid cells indicate a percentage greater than 0, white grid cells indicate no tipping points.

In the more moderate SLR scenarios 3–6 (note that 4–9 roughly correspond to the IPCC SROCC 'likely range', Figure SI 3) SETPs may still occur in the City Centre, but the pattern is more varied. In the proactive strategy, no SETPs occur. In the other three strategies, SETPs only occur in the boundedly rational housing market where flood risk perceptions (rather than rational risk) determine house prices. In other words: in an economic rational market, no busts of house prices are expected under moderate SLR scenarios. In these markets, the risk only gradually increases and is reduced by adaptation before becoming substantial. However, if housing prices depend on more strongly fluctuating risk perceptions, such as in the boundedly rational market, rapid drops in house prices occur. The difference between these two housing markets can be seen in Fig. 5, panel 4B.

In the low-end SLR scenarios (7–9), no tipping points occur in the City Centre. But in the unembanked area A, they do occur, even before the year 2100 (Figure SI 17).

4.5. Step 5: Impact assessment

What could be the impact of a house price collapse in the small unembanked area A? If market forces dominate, the residents who can afford it will sell their properties and relocate to safer grounds whereas low-income and less risk-adverse residents might be attracted by the lower prices. This could trap low-income households in the area until the point is reached where the area becomes unliveable (de Koning and Filatova, 2020). More likely, the government will intervene, by forcing retreat or entire redevelopment of the area.

A house price collapse in the (much larger) city centre (area B) would most likely hamper real estate investments and reduce regional economic growth (Beltratti and Morana, 2010). It could also increase demand and inflate prices of property on safer grounds. To the extent that the collapse was driven by unrealistically high risk perceptions, market forces and trust may restore prices to rational levels.

These urban-scale impacts are likely to spill over to the national scale. Especially in the Dutch situation, where the flood-prone Randstad megapolis hosts about half of the population and GDP (CBS Statistics Netherlands, 2021b), any serious concern about flood safety will trigger large debate. Large parts of the Netherlands currently are less protected than Rotterdam because of smaller economic activity (Kind, 2014). This observation exemplifies how assessment of the wider system context in step 5 could trigger the feedback loop to step 1: when reflecting on the impact of an SETP for Rotterdam, one discovers that this cannot be seen in isolation from the spatial planning strategy of the entire country and one may repeat the analysis with wider system boundaries.

4.6. Step 6: Action identification

The experimental design included the widely observed *protect* strategy of incremental dike heightening measures to protect the city centre. The proactiveness by which this strategy was carried out was implemented as an external policy lever, so that the timing of individual measure implementation could be endogenously simulated. The results show that the *protect* strategy is mainly effective in moderate SLR scenarios and when it is carried out proactively, otherwise SETPs might occur. Monitoring of SLR, flood events and changes in house prices could provide early warning of an approaching SETP. Besides this incremental *protect* strategy, one could also *accommodate* to make the city less vulnerable to floods, *advance* by expanding and elevating the city seaward, or *retreat* by relocating houses to safer grounds (Mach et al., 2019; Oppenheimer et al., 2019).

In the unembanked residential area, one could for example *accommodate* by implementing loss-reduction measures, which reduce damage by 70 % for inundations < 1 m; the study of Haer et al. (2017) shows that this is an effective means to cope with moderate SLR scenarios. More drastic *accommodation* would be to elevate entire residential areas. The *advance* response is already used in the newly reclaimed areas of the Port of Rotterdam, which are constructed 5 m above current mean sea level (Boer et al., 2009). By using the feedback loop from step six to step two, one could add the flood proofing and elevation actions as a new policy levers and explore its impact on SETP-occurrence.

On the national scale, a *retreat* strategy could be the relocation of city residents to higher grounds in the east of the Netherlands. Especially for areas with small socio-economic importance, managed retreat could be more than a mere left-over strategy, provided it is embedded in sound national spatial planning (Mach et al., 2019; Siders et al., 2019). On the international scale, a very drastic *advance* action would be the construction of the Northern European Enclosure Dam (Groeskamp and Kjellsson, 2020) enabling water level control or even complete reclamation of the North Sea. Such transformative measures can be considered socio-economic policy tipping points (Herrfahrdt-Pähle et al., 2020; van Ginkel et al., 2020), in terms of *transformational response* SETPs (box 1), which could be added as a policy lever to the current model set-up in step 2. Evidently, it would be more meaningful to do such analysis with wider system boundaries, one could then also decide to include the decision-making process on local retreat as endogenous system behaviour, and study it as an impact SETP.

4.7. Step 7: Decision support

What policy advice can be given based on the SETP-analysis? First, a very proactive flood risk management strategy can already avoid most of the SETPs. Second, the implementation times of flood protection measures need to be sufficiently short to keep pace with the rate of sea level rise. Under high-end sea level rise, even a very proactive management strategy might fail if the measure implementation time is too long. This is alarming because at present, technical challenges and complex coastal management frameworks often cause long implementation times. Third, management of flood risk perception is very important: having adequate flood protection is one thing, convincing the public that it is adequate is another. When all buyers and sellers on the house market are aware of

gradual changes in flood risk, large fluctuations in risk perceptions and related house price booms and busts may be avoided. Whereas the other insights also emerge from other studies as good management practices, this finding most uniquely follows from the SETP-perspective taken in study. It highlights that for avoiding SETPs, socio-economic processes like opinion dynamics and risk perceptions could be as important as technical adaptation measures (cf. Smith et al., 2022). Fourth, monitoring of sea level rise and ice sheet dynamics is crucial to detect any possible acceleration of SLR early. In the most extreme scenarios, there is a risk that the *protect* strategy proves to be a lock-in strategy which ultimately fails. Then a possible switch to a different coastal adaptation strategy such as *advance, accommodate,* or *retreat* should be anticipated as soon as possible, because such fundamental policy shifts may require much longer implementation times (Termeer et al., 2017).

5. Discussion

The aim of this study was to present a stepwise approach that identifies under what conditions SETPs may occur and guides policy aimed at preventing adverse SETPs or supporting positive SETPs. With the help of a stylized model we explored many possible futures of a coastal city and did identify SETPs. Sudden house price collapses occur when extreme sea level rise is compounded by: reactive flood risk management, risk perceptions dominating house price dynamics, or long implementation times of flood protection measures. A frequent cause is when during several decades the actual flood risk gradually increases to substantial levels without triggering action, because by chance there are no major storm surges. When a trigger event eventually occurs, this causes a large drop in house price, because the substantial flood risk is suddenly and amplified (due to risk overestimation) discounted for in the house price. This eventually triggers large flood protection measures, which take time to implement. In the most extreme SLR scenario's, the rate of sea level rise exceeds the speed and magnitude at which the government can respond, and the house prices remain unstable.

These findings indeed help to inform policy aiming at preventing SETPs; Section 4.7 presented four policy recommendations for coastal cities like Rotterdam. Either a reactive 'wait-and-see' policy, a failure to manage flood risk perceptions, or a delayed implementation of a crucial flood protection measure may cause an SETP in high-end SLR scenarios. If this can happen to a relatively prosperous city with very high current protection levels like Rotterdam, we may expect that in other coastal cities, SETPs may happen earlier and also in less extreme SLR scenarios because of weaker institutional settings and less resources for adequate response (cf. Hinkel et al., 2018; van Ginkel et al., 2018). This confirms findings of other studies that stress the need for a proactive, adaptive flood risk management strategy which monitors important signpost and responds timely (e.g. Haasnoot et al., 2018).

Given the stylized relations in the model, the case study results should be interpreted with caution. The SLR-scenarios 1 and 2 follow the \geq 95 % tail of the probability distribution (Figure SI 3), meaning that even when accounting for ice sheet instabilities, these have low likelihood (Bamber et al., 2019). Since the house markets are archetypes of actual dynamics, the price fluctuations under extreme conditions are rather large. Real-world house price fluctuations can also be large: in 1999, the house price in Amsterdam increased by 19 % (CBS Statistics Netherlands, 2021a) and from 2006 to 2009, nationwide property prices in the US dropped by 38 % (Nneji et al., 2013). The policy responses are generalized features of possible dynamic adaptive strategies, whereas reality is more nuanced.

However, our objective was to illustrate the stepwise approach, for which our model is suitable due to three characteristics. First, the short runtime allows for the exploration of many possible futures, uncertainties and policies. Second, the model relations cause non-linear behaviour under some conditions giving it the potential to exhibit tipping points. Third, the model dynamically adapts over time, which is an indispensable feature to represent a highly adaptive socio-economic system on longer timescales. For other applications of the stepwise approach, models with these three characteristics are suitable, such as some agent-based, system-dynamic or graph-theoretic models (Filatova et al., 2016).

How can the model approach be applied to other coastal zones? The current model structure contains many generic aspects applicable to other areas, such as the built-in sea level rise scenarios, a storm surge generator, a flood damage and risk calculator, dynamic adaptive dike heightening strategies, as well as a risk perception, house price, and SETP-identification module. These can easily be adjusted to the local context. At the same time, some model relations will need to be updated to represent the myriad of complex dynamic adaptive processes that dominate in different socio-economic, institutional and biophysical contexts around the globe. For example, coastal adaptation in the USA relies heavily on adaptation by households, firms and communities, and not solely on government interventions. Also, the role of insurance is highly relevant (Ruig et al., 2022). To mimic house-price and adaptive dynamics, the model needs to be downscaled to individual household actors, which could build on studies like De Ruig et al. (2022) for New York City. In a city like Miami, there can be technological limits - a highly permeable underground - that constrain the dike heightening solution and need to be accounted for. A city like Beira (Mozambique) suffers from large financial barriers due to a very poor credit rating, which means that a strategy cannot be based on cost-benefit analysis alone, but also relies heavily on the availability of financial means. Another aspect is poor maintenance of coastal flood protection infrastructure in times of economic or institutional downturn, which could be represented with enhanced failure probabilities of the flood protection infrastructure. Furthermore, the range of management strategies can be extended to *retreat, accommodate,* and *advance,* to complement the current *protect* strategy, as discussed in section 4.5 and 4.6.

In future research, the stepwise approach helps to get a grip on the core elements, assumptions, definitions and methodologies of research on policy relevant tipping points. The advantage of using the steps is that they force reflection on important SETP-characteristics. First, SETPs typically occur on specific spatial, temporal and sectoral scales and are less distinct on other scales. The analysis therefore initially focusses on a strictly delineated scale, whereas possible spill-overs to other scales are later examined in the impact analysis step. A related characteristic is that SETPs may unevenly impact different stakeholder groups, because socio-economic reconfiguration typically has winners and losers. Transparency about the reflected perspective may avoid controversy

about tipping point terminology and examples. A second advantage is that one does not have to decide between these possible conflicting views on how the system works, because multiple possible system relations (such as our two types of house markets) can be represented. Third, SETPs seem rare phenomena resulting from specific compound biophysical and socio-economic events, located somewhere in the large uncertainty space. The approach transparently describes relevant uncertainties and provides a systematic procedure to identify the (rare) points of interest. Fourth, the approach delivers a concrete policy advice by identification of adaptation and mitigation actions, and recommendation of key monitoring variables.

Although our focus has been quantitative, the stepwise approach can also be used for qualitative assessments. In step 3, a narrative would operationalise the SETP by describing two more or less stable, fundamentally different states at both sides of a critical threshold, and how a rapid shift between these states may occur. Such descriptions may be supported from archaeological evidence (e.g. Tainter, 1988), scientific descriptions of socio-economic dynamics (e.g. McLeman, 2017) or from expert and stakeholder consultation (e.g. Olsthoorn et al., 2008). In step 4, a set of plausible qualitive future storylines would be developed, based on literature review or stakeholder interviews/workshops (e.g. Jack et al., 2020; Lopez et al., 2019; Wilby and Dessai, 2010; van Ginkel et al., 2020), that indicate under what conditions tipping points occur in the range of possible futures (cf. Otto et al., 2019). One pitfall of a qualitative approach is that anecdotical evidence or unrealistic views of particular stakeholders start dominating the storylines; in an earlier qualitative exploration of tipping points in the Dutch Delta, stakeholders had diverging and conflicting views on what could possibly happen (van Ginkel et al., 2020). A complementary model-based approach can help to assess the plausibility of storylines and to systematically explore large outcome spaces, which are hard to comprehend without computational support.

In-between qualitative and quantitative, one can also take a 'metric-over-critical-threshold' approach. Here, the metric itself does not show non-linear behaviour, but is known to have a critical threshold value, at which a transition between two stable system states occurs (Barnard et al., 2021). An example threshold is the 100-day rule used for analysing climate change impacts to ski resorts (Damm et al., 2014), indicating the minimum number of favourable snow days per year for a ski resort to be profitable. If snow conditions repetitively drop below this threshold, the ski resort will financially collapse (Vaghefi et al., 2021). The upside of the metric-over-threshold approach is that the complex non-linear behaviour around the tipping point does not have to be modelled. The downside is the a priori assumption that the threshold indicates an SETP; this assumption may be false in parts of the uncertainty space.

The fully quantitative approach also has specific challenges. The first is developing an algorithm that soundly detects SETPs. This challenge is more than just technical, because it highlights a tension between a strict system-dynamic understanding vs a more policy relevant understanding of tipping points (see Van Ginkel et al., 2022 for an example in transport networks). For our case, the situation where the house price abruptly dropped after a stable period and kept fluctuating without reaching a new equilibrium, does not meet all three SETP criteria, but is nevertheless policy relevant. We conclude that typically, the set of policy relevant points is larger than the set of perfect SETP examples, because large metric fluctuations are often policy relevant, even if they are not abrupt or persistent. Second, one needs to verify if initial model relations and boundary conditions are still valid when the system has tipped. By definition, the state after the SETP fundamentally differs from the state before. Therefore, other socio-economic mechanisms than initially modelled may dominate the model when the tipping point. At such below-market-value prices, the residential areas might also become available for redevelopment into industrial or port areas, or nature development. These price-stabilizing mechanisms are not incorporated in the model, but even if they would be included, the new equilibrium state would be fundamentally different, meaning the SETP definition is met. The value of the approach lies therefore mainly in detecting SETPs rather than precise forecasting how the new state will look like.

6. Conclusion

Climate change may cause socio-economic tipping points, which are uncertain but policy relevant. This study presents a stepwise approach that identifies under what conditions SETPs may occur and guides policy aimed at preventing adverse SETPs or supporting positive SETPs. This fills a niche in existing literature which mainly focussed on biophysical non-linearities and lacked a policy-relevant approach that accounts for many uncertainties. It adds to DMDU-literature an approach that explicitly accounts for tipping points.

The stepwise approach can range from qualitative, with stakeholder narratives, to quantitative, with models and exploratory modelling tools. Qualitative assessment is a resource efficient method to explore many mechanisms and quantitative assessment is important to verify these mechanisms. In a quantitative assessment, existing DMDU techniques such as scenario discovery are extended with a tipping point identification algorithm to investigate output timeseries. We found that the set of policy relevant points is typically larger than only the tipping points according to a strict system-dynamic understanding.

The case study suggests that even in a city that currently has very high protection levels (such as Rotterdam), house price collapses may occur if in a high-end sea level rise scenario, the government fails to act fast and proactively. Collapses result from sudden increases of public flood risk perceptions; managing these perceptions is therefore one of our policy recommendations. In the Dutch context, structural solutions to the most extreme SLR scenarios are found on larger spatial levels than the urban scale of this study. When monitoring reveals that a high-end SLR scenario is approaching, an intentional *transformative response SETP* towards an *accommodate* or *retreat* strategy could be triggered.

A natural progression of this study are new applications of the model to the long-term adaptive dynamics of other coastal areas around the world. It can investigate tipping points under different biophysical, socio-economic and institutional settings. The model code is open source and can be easily extended and tailored to the local context (see Data and code availability).

Future research can benefit from applying the stepwise approach; see Van Ginkel et al. (2022) for a recent application to transport networks. The system delineation, system description and tipping point operationalisation step provide insight into the assumptions

underlying the assessment, which clarifies the applicability, validity and scope of the results. The exploratory modelling step systematically explores the relevant uncertainties which is imperative because SETPs are typically rare, compound events. The impact assessment revisits the choice of system scale, which can be decisive for finding tipping points or not. The steps eventually lead to a concrete policy advice, which moves a study beyond a mere analysis of SETPs as an interesting system-dynamic phenomenon, towards a concept that provides perspective of action for policy makers.

Application of the stepwise approach to cases where environmental change occurs in combination with non-linear socio-economic dynamics will contribute to a better understanding of socio-economic tipping points and will support policies to mitigate or adapt to them.

Data and code availability

All model code and data, also for reproduction of Figs. 5-7, can be found on GitHub through this https://doi.org/10.5281/zenodo. 6782856.

Funding

This paper has received funding from the European Union's Horizon 2020 research and innovation programme under grant agreement No 776479 for the project CO-designing the Assessment of Climate CHange costs. https://www.coacch.eu/.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

All code and data can be found in the Supplementary Infomation and on the Github repository.

Appendix A. Supplementary data

Supplementary data to this article can be found online at https://doi.org/10.1016/j.crm.2022.100445.

References

Aerts, J.C.J.H., 2020. Integrating agent-based approaches with flood risk models: A review and perspective. Water Secur. 11, 100076 https://doi.org/10.1016/j. wasec.2020.100076.

Andersen, T., Carstensen, J., Hernández-García, E., Duarte, C.M., 2009. Ecological thresholds and regime shifts: approaches to identification. Trends Ecol. Evol. 24, 49–57. https://doi.org/10.1016/j.tree.2008.07.014.

- Bamber, J.L., Oppenheimer, M., Kopp, R.E., Aspinall, W.P., Cooke, R.M., 2019. Ice sheet contributions to future sea-level rise from structured expert judgment. Proc. Natl. Acad. Sci. U. S. A. 166, 11195–11200. https://doi.org/10.1073/pnas.1817205116.
- Barnard, P.L., Dugan, J.E., Page, H.M., Wood, N.J., Hart, J.A.F., Cayan, D.R., Erikson, L.H., Hubbard, D.M., Myers, M.R., Melack, J.M., Iacobellis, S.F., 2021. Multiple climate change-driven tipping points for coastal systems. Sci. Rep. 11, 1–13. https://doi.org/10.1038/s41598-021-94942-7.
- Beltrán, A., Maddison, D., Elliott, R.J.R., 2018. Is Flood Risk Capitalised Into Property Values? Ecol. Econ. 146, 668–685. https://doi.org/10.1016/j.
- Beltratti, A., Morana, C., 2010. International house prices and macroeconomic fluctuations. J. Bank. Financ. 34, 533–545. https://doi.org/10.1016/j. ibankfin.2009.08.020.
- Berrang-Ford, L., Siders, A.R., Lesnikowski, A., Fischer, A.P., Callaghan, M.W., Haddaway, N.R., Mach, K.J., Araos, M., Shah, M.A.R., Wannewitz, M., Doshi, D., Leiter, T., Matavel, C., Musah-Surugu, J.I., Wong-Parodi, G., Antwi-Agyei, P., Ajibade, I., Chauhan, N., Kakenmaster, W., Grady, C., Chalastani, V.I., Jagannathan, K., Galappaththi, E.K., Sitati, A., Scarpa, G., Totin, E., Davis, K., Hamilton, N.C., Kirchhoff, C.J., Kumar, P., Pentz, B., Simpson, N.P., Theokritoff, E., Deryng, D., Reckien, D., Zavaleta-Cortijo, C., Ulibarri, N., Segnon, A.C., Khavhagali, V., Shang, Y., Zvobgo, L., Zommers, Z., Xu, J., Williams, P.A., Canosa, I.V., van Maanen, N., van Bavel, B., van Aalst, M., Turek-Hankins, L.L., Trivedi, H., Trisos, C.H., Thomas, A., Thakur, S., Templeman, S., Stringer, L.C., Sotnik, G., Sjostrom, K.D., Singh, C., Sina, M.Z., Shukla, R., Sardans, J., Salubi, E.A., Safaee Chalkasra, L.S., Ruiz-Díaz, R., Richards, C., Pokharel, P., Petzold, J., Penuelas, J., Pelaez Avila, J., Murillo, J.B.P., Ouni, S., Niemann, J., Nielsen, M., New, M., Nayna Schwerdtle, P., Nagle Alverio, G., Mullin, C.A., Mullenite, J., Mosurska, A., Morecroft, M.D., Minx, J.C., Maskell, G., Nunbogu, A.M., Magnan, A.K., Lwasa, S., Lukas-Sithole, M., Lissner, T., Lilford, O., Koller, S.F., Jurjonas, M., Joe, E.T., Huynh, L.T.M., Hill, A., Hernandez, R.R., Hegde, G., Hawxwell, T., Harper, S., Harden, A., Haasnoot, M., Gilmore, E.A., Gichuki, L., Gatt, A., Garschagen, M., Ford, J.D., Forbes, A., Farrell, A.D., Enquist, C.A.F., Elliott, S., Duncan, E., Coughlan de Perez, E., Coggins, S., Chen, T., Gambell, D., Browne, K.E., Bowen, K.J., Biesbroek, R., Bhat, I.D., Bezner Kerr, R., Barte, S., Baker, E., Austin, S.E., Arotoma-Rojas, I., Anderson, C., Ajaz, W., Agrawal, T., Abu, T.Z., 2021. A systematic global stocktake of evidence on human adaptation to climate change. Nat. Clim. Chang. 11, 989–1000. https://doi.org/10.1038/s41558-021-01170-y.
- Biggs, R., Peterson, G.D., Rocha, J.C., 2018. The regime shifts database: A framework for analyzing regime shifts in social-ecological systems. Ecol. Soc. 23 https://doi. org/10.5751/ES-10264-230309.
- Bin, O., Landry, C.E., 2013. Changes in implicit flood risk premiums: Empirical evidence from the housing market. J. Environ. Econ. Manage. 65, 361–376. https://doi.org/10.1016/j.jeem.2012.12.002.
- Bregman, R., 2020. This is what climate change means if your country is below sea level. In: The Correspondent, 24 September 2020. Retrieved from: https:// thecorrespondent.com/685/this-is-what-climate-change-means-if-your-country-is-below-sea-level/90630422410-d543c5e9, latest access 12 March 2021.
- Boer, S., Konter, J., Meijer, A.K., 2009. Rational risk based design method for flood protection Maasvlakte 2, Port of Rotterdam in: Coastal Engineering 2008. World Scientific Publishing Company, pp. 4289–4301. 10.1142/9789814277426_0356.

- Bordner, A.S., Ferguson, C.E., Ortolano, L., 2020. Colonial dynamics limit climate adaptation in Oceania: Perspectives from the Marshall Islands. Glob. Environ. Chang. 61, 102054 https://doi.org/10.1016/j.gloenvcha.2020.102054.
- Bosker, M., Garretsen, H., Marlet, G., Van Woerkens, C., 2019. Nether lands: Evidence on the price and perception of rare natural disasters. J. Eur. Econ. Assoc. 17, 413–453. https://doi.org/10.1093/jeea/jvy002.
- Botzen, W.J.W., Aerts, J.C.J.H., van den Bergh, J.C.J.M., 2009. Dependence of flood risk perceptions on socioeconomic and objective risk factors. Water Resour. Res. 45, 1–15. https://doi.org/10.1029/2009WR007743.
- Brown, C., Ghile, Y., Laverty, M., Li, K., 2012. Decision scaling: Linking bottom-up vulnerability analysis with climate projections in the water sector. Water Resour. Res. 48, 1–12. https://doi.org/10.1029/2011WR011212.
- Brown, S., Nicholls, R.J., Goodwin, P., Haigh, I.D., Lincke, D., Vafeidis, A.T., Hinkel, J., 2018. Quantifying Land and People Exposed to Sea-Level Rise with No Mitigation and 1.5°C and 2.0°C Rise in Global Temperatures to Year 2300. Earth's Futur. 6, 583–600. https://doi.org/10.1002/2017EF000738.
- Bryant, B.P., Lempert, R.J., 2010. Thinking inside the box: A participatory, computer-assisted approach to scenario discovery. Technol. Forecast. Soc. Change 77, 34–49. https://doi.org/10.1016/i.techfore.2009.08.002.
- Burch, S., Mitchell, C., Berbes-Blazquez, M., Wandel, J., 2017. Tipping Toward Transformation: Progress, Patterns and Potential for Climate Change Adaptation in the Global South. J. Extrem. Events 4, 1750003. https://doi.org/10.1142/S2345737617500038.
- Caloia, F., Jansen, D., 2021. Flood risk and financial stability: evidence from a stress test for the Netherlands. Retrieved from De Nederlandsche Bank, www.dnb.nl, latest access 4 november 2021. DNB working paper 730.

Carleton, T.A., Hsiang, S.M., 2016. Social and economic impacts of climate. Science (80-.). 353 https://doi.org/10.1126/science.aad9837.

- Carstensen, J., Weydmann, A., 2012. Tipping Points in the Arctic: Eyeballing or Statistical Significance? Ambio 41, 34–43. https://doi.org/10.1007/s13280-011-0223-8.
- CBS Statistics Netherlands, 2021A. StatLine: Bestaande koopwoningen; verkoopprijzen. https://opendata.cbs.nl/statline/#/CBS/nl/dataset/83913NED/table? ts=1611669293798. Last access: 11 March 2021.
- CBS Statistics Netherlands, 2021B. StatLine: Regionale kerncijfers; nationale rekeningen. https://opendata.cbs.nl/statline/#/CBS/nl/dataset/84432NED/table? ts=1615461250958. Last access: 24 February 2021.
- Damm, A., Köberl, J., Prettenthaler, F., 2014. Does artificial snow production pay under future climate conditions?-A case study for a vulnerable ski area in Austria. Tour. Manag. 43, 8–21. https://doi.org/10.1016/j.tourman.2014.01.009.
- de Koning, K., Filatova, T., Need, A., Bin, O., 2019. Avoiding or mitigating flooding: Bottom-up drivers of urban resilience to climate change in the USA. Glob. Environ. Chang. 59, 101981 https://doi.org/10.1016/j.gloenvcha.2019.101981.
- de Koning, K., Filatova, T., 2020. Repetitive floods intensify outmigration and climate gentrification in coastal cities. Environ. Res. Lett. 15, 034008 https://doi.org/ 10.1088/1748-9326/ab6668.
- de Ruig, L.T., Haer, T., de Moel, H., Orton, P., Botzen, W.J.W., Aerts, J.C.J.H., 2022. An agent-based model for evaluating reforms of the National Flood Insurance Program: A benchmarked model applied to Jamaica Bay. NYC. Risk Anal. 1–18 https://doi.org/10.1111/risa.13905.
- Dietz, S., Rising, J., Stoerk, T., Wagner, G., 2021. Economic impacts of tipping points in the climate system. Proc. Natl. Acad. Sci. U. S. A. 118 https://doi.org/ 10.1073/pnas.2103081118.
- Fedele, G., Donatti, C.I., Harvey, C.A., Hannah, L., Hole, D.G., 2019. Transformative adaptation to climate change for sustainable social-ecological systems. Environ. Sci. Policy 101, 116–125. https://doi.org/10.1016/j.envsci.2019.07.001.
- Filatova, T., Polhill, J.G., van Ewijk, S., 2016. Regime shifts in coupled socio-environmental systems: Review of modelling challenges and approaches. Environ. Model. Softw. 75. 333–347. https://doi.org/10.1016/j.envsoft.2015.04.003.
- Garschagen, M., Solecki, W., 2017. Editorial Tipping Points in Adaptive Capacity and Adaptation Processes. J. Extrem. Events 04, 1702002. https://doi.org/ 10.1142/S234573761702002X.

Gladwell, M., 2000. The tipping point: How little things can make a big difference. Little Brown, London.

- Groeskamp, S., Kjellsson, J., 2020. NEED: The Northern European Enclosure Dam for if Climate Change Mitigation Fails. Bull. Am. Meteorol. Soc. 101, E1174–E1189. https://doi.org/10.1175/BAMS-D-19-0145.1.
- Groves, D.G., Lempert, R.J., 2007. A new analytic method for finding policy-relevant scenarios. Glob. Environ. Chang. 17, 73–85. https://doi.org/10.1016/j.gloenvcha.2006.11.006.
- Haasnoot, M., Kwadijk, J., van Alphen, J., Le Bars, D., van den Hurk, B., Diermanse, F., van der Spek, A., Essink, G.O., Delsman, J., Mens, M., 2020. Adaptation to uncertain sea-level rise; how uncertainty in Antarctic mass-loss impacts the coastal adaptation strategy of the Netherlands. Environ. Res. Lett. 15, 034007 https:// doi.org/10.1088/1748-9326/ab666c.
- Haasnoot, M., van 't Klooster, S., van Alphen, J.,, 2018. Designing a monitoring system to detect signals to adapt to uncertain climate change. Glob. Environ. Chang. 52, 273–285. https://doi.org/10.1016/j.gloenvcha.2018.08.003.
- Hadjimichael, A., Gold, D., Hadka, D., Reed, P., 2020. Rhodium: Python library for many-objective robust decision making and exploratory modeling. J. Open Res. Softw. 8, 1–10. https://doi.org/10.5334/JORS.293.
- Haer, T., Botzen, W.J.W., de Moel, H., Aerts, J.C.J.H., 2017. Integrating Household Risk Mitigation Behavior in Flood Risk Analysis: An Agent-Based Model Approach. Risk Anal. 37, 1977–1992. https://doi.org/10.1111/risa.12740.
- Hallegatte, S., Green, C., Nicholls, R.J., Corfee-Morlot, J., 2013. Future flood losses in major coastal cities. Nat. Clim. Chang. 3, 802–806. https://doi.org/10.1038/ nclimate1979.
- Herrfahrdt-Pähle, E., Schlüter, M., Olsson, P., Folke, C., Gelcich, S., Pahl-Wostl, C., 2020. Sustainability transformations: socio-political shocks as opportunities for governance transitions. Glob. Environ. Chang. 63, 102097 https://doi.org/10.1016/j.gloenvcha.2020.102097.
- Hinkel, J., Aerts, J.C.J.H., Brown, S., Jiménez, J.A., Lincke, D., Nicholls, R.J., Scussolini, P., Sanchez-Arcilla, A., Vafeidis, A., Addo, K.A., 2018. The ability of societies to adapt to twenty-first-century sea-level rise. Nat. Clim. Chang. 8, 570–578. https://doi.org/10.1038/s41558-018-0176-z.
- Horton, B.P., Khan, N.S., Cahill, N., Lee, J.S.H., Shaw, T.A., Garner, A.J., Kemp, A.C., Engelhart, S.E., Rahmstorf, S., 2020. Estimating global mean sea-level rise and its uncertainties by 2100 and 2300 from an expert survey. npj Clim. Atmos. Sci. 3, 1–8. https://doi.org/10.1038/s41612-020-0121-5.
- Hossain, M.S., Dearing, J.A., Eigenbrod, F., Johnson, F.A., 2017. Operationalizing safe operating space for regional social-ecological systems. Sci. Total Environ. 584–585, 673–682. https://doi.org/10.1016/j.scitotenv.2017.01.095.
- IPCC, 2014: Climate Change 2014: Impacts, Adaptation, and Vulnerability. Part A: Global and Sectoral Aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Field, C.B., V.R. Barros, D.J. Dokken, K.J. Mach, M.D. Mastrandrea, T.E. Bilir, M. Chatterjee, K.L. Ebi, Y.O. Estrada, R.C. Genova, B. Girma, E.S. Kissel, A.N. Levy, S. MacCracken, P.R. Mastrandrea, and L.L. White (eds.)]. Cambridge University Press, Cambridge, United Kingdom and New York, NY, USA, 1132 pp.
- Jack, C.D., Jones, R., Burgin, L., Daron, J., 2020. Climate risk narratives: An iterative reflective process for co-producing and integrating climate knowledge. Clim. Risk Manag. 29 https://doi.org/10.1016/j.crm.2020.100239.
- Kabir, M.E., Serrao-Neumann, S., Davey, P., Hossain, M., Alam, M.T., 2018. Drivers and temporality of internal migration in the context of slow-onset natural hazards: Insights from north-west rural Bangladesh. Int. J. Disaster Risk Reduct. 31, 617–626. https://doi.org/10.1016/j.ijdrr.2018.06.010.
- Karstens, S.A.M., Bots, P.W.G., Slinger, J.H., 2007. Spatial boundary choice and the views of different actors. Environ. Impact Assess. Rev. 27, 386–407. https://doi. org/10.1016/j.eiar.2007.02.002.
- Kates, R.W., Travis, W.R., Wilbanks, T.J., 2012. Transformational adaptation when incremental adaptations to climate change are insufficient. Proc. Natl. Acad. Sci. 109, 7156–7161. https://doi.org/10.1073/pnas.1115521109.
- Kind, J.M., 2014. Economically efficient flood protection standards for the Netherlands. J. Flood Risk Manag. 7, 103–117. https://doi.org/10.1111/jfr3.12026.
- Kopp, R.E., Shwom, R.L., Wagner, G., Yuan, J., 2016. Tipping elements and climate-economic shocks: Pathways toward integrated assessment. Earth's Futur. 4, 346–372. https://doi.org/10.1002/2016EF000362.

- Kwadijk, J.C.J., Haasnoot, M., Mulder, J.P.M., Hoogvliet, M.M.C., Jeuken, A.B.M., van der Krogt, R.A.A., van Oostrom, N.G.C., Schelfhout, H.A., van Velzen, E.H., van Waveren, H., de Wit, M.J.M., 2010. Using adaptation tipping points to prepare for climate change and sea level rise: a case study in the Netherlands. Wiley Interdiscip. Rev. Clim. Chang. 1, 729–740. https://doi.org/10.1002/wcc.64.
- Kwakkel, J.H., 2017. The Exploratory Modeling Workbench: An open source toolkit for exploratory modeling, scenario discovery, and (multi-objective) robust decision making. Environ. Model. Softw. 96, 239–250. https://doi.org/10.1016/j.envsoft.2017.06.054.
- Le Bars, D., Drijfhout, S., de Vries, H., 2017. A high-end sea level rise probabilistic projection including rapid Antarctic ice sheet mass loss. Environ. Res. Lett. 12, 044013 https://doi.org/10.1088/1748-9326/aa6512.

Lemoine, D., Traeger, C.P., 2016. Economics of tipping the climate dominoes. Nat. Clim. Chang. 6, 514–519. https://doi.org/10.1038/nclimate2902.

- Lempert, R.J., Popper, S.W., Bankes, S.C., 2005. Shaping the Next One Hundred Years: New Methods for Quantitative, Long-Term Policy Analysis. Lenton, T.M., Held, H., Kriegler, E., Hall, J.W., Lucht, W., Rahmstorf, S., Schellnhuber, H.J., 2008. Tipping elements in the Earth's climate system. Proc. Natl. Acad.
- Sci. U. S. A. 105, 1786–1793. https://doi.org/10.1073/pnas.0705414105.
 Lontzek, T.S., Cai, Y., Judd, K.L., Lenton, T.M., 2015. Stochastic integrated assessment of climate tipping points indicates the need for strict climate policy. Nat. Clim. Chang. 5, 441–444. https://doi.org/10.1038/nclimate2570.
- Lopez, J.M.R., Tielbörger, K., Claus, C., Fröhlich, C., Gramberger, M., Scheffran, J., 2019. A transdisciplinary approach to identifying transboundary tipping points in a contentious area: Experiences from across the Jordan River region. Sustain. 11, 1–20. https://doi.org/10.3390/su11041184.
- Mach, K.J., Kraan, C.M., Hino, M., Siders, A.R., Johnston, E.M., Field, C.B., 2019. Managed retreat through voluntary buyouts of flood-prone properties. Sci. Adv. 5, 1–10. https://doi.org/10.1126/sciadv.aax8995.
- Magnan, A.K., Duvat, V.K.E., 2020. Towards adaptation pathways for atoll islands. Insights from the Maldives. Reg. Environ. Chang. 20 https://doi.org/10.1007/s10113-020-01691-w.
- Maier, R., Chakraborty, S., Steininger, K.W., Mandel, A, 2020. Report with literature review advancing the state of the art on research on tipping points in economics. Deliverable 4.1 of the Tipping-Plus project. Retrieved from www.tipping-plus.eu, last accessed on 2 November 2021.
- Marchau, V.A.W.J., Walker, W.E., Bloemen, P.J.T.M., Popper, S.W., 2019. Decision Making under Deep Uncertainty. Springer International Publishing, Cham. https:// doi.org/10.1007/978-3-030-05252-2.
- McLeman, R., 2017. Thresholds in climate migration. Popul. Environ. 1-20 https://doi.org/10.1007/s11111-017-0290-2.
- McNamara, D.E., Keeler, A., 2013. A coupled physical and economic model of the response of coastal real estate to climate risk. Nat. Clim. Chang. 3, 559–562. https://doi.org/10.1038/nclimate1826.
- Milkoreit, M., Hodbod, J., Baggio, J., Benessaiah, K., Calderón-Contreras, R., Donges, J.F., Mathias, J.-D., Rocha, J.C., Schoon, M., Werners, S.E., 2018. Defining tipping points for social-ecological systems scholarship—an interdisciplinary literature review. Environ. Res. Lett. 13, 033005 https://doi.org/10.1088/1748-9326/aaaa75.
- Moallemi, E.A., Kwakkel, J., de Haan, F.J., Bryan, B.A., 2020. Exploratory modeling for analyzing coupled human-natural systems under uncertainty. Glob. Environ. Chang. 65, 102186 https://doi.org/10.1016/j.gloenvcha.2020.102186.
- Neumann, B., Vafeidis, A.T., Zimmermann, J., Nicholls, R.J., 2015. Future Coastal Population Growth and Exposure to Sea-Level Rise and Coastal Flooding A Global Assessment. PLoS One 10, e0118571.
- Nneji, O., Brooks, C., Ward, C.W.R., 2013. House price dynamics and their reaction to macroeconomic changes. Econ. Model. 32, 172–178. https://doi.org/10.1016/j. econmod.2013.02.007.
- Nordhaus, W., 2018. Evolution of modeling of the economics of global warming: changes in the DICE model, 1992–2017. Clim. Change 148, 623–640. https://doi.org/10.1007/s10584-018-2218-y.
- Olsthoorn, X., Van Der Werff, P., Bouwer, L.M., Huitema, D., 2008. Neo-Atlantis: The Netherlands under a 5-m sea level rise. Clim. Change 91, 103–122. https://doi.org/10.1007/s10584-008-9423-z.
- Oppenheimer, M., B.C. Glavovic, J. Hinkel, R. van de Wal, A.K. Magnan, A. Abd-Elgawad, R. Cai, M. Cifuentes-Jara, R.M. DeConto, T. Ghosh, J. Hay, F. Isla, B. Marzeion, B. Meyssignac, and Z. Sebesvari, 2019: Sea Level Rise and Implications for Low-Lying Islands, Coasts and Communities. In: IPCC Special Report on the Ocean and Cryosphere in a Changing Climate [H.-O. Pörtner, D.C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, K. Mintenbeck, A. Alegría, M. Nicolai, A. Okem, J. Petzold, B. Rama, N.M. Weyer (eds.)].
- Otto, I.M., Donges, J.F., Bhowmik, A., Cremades, R., Lucht, W., Rockström, J., Allerberger, F., Doe, S., Hewitt, R., Lenferna, A., McCaffrey, M., Morán, N., van

Vuuren, D.P., Schellnhuber, H.J., 2019. Social tipping dynamics for stabilizing Earth's climate by 2050. Pnas. https://doi.org/10.1073/pnas.1900577117.

- Quandt, R.E., 1958. The Estimation of the Parameters of a Linear Regression System Obeying Two Separate Regimes. J. Am. Stat. Assoc. 53, 873–880. https://doi.org/ 10.1080/01621459.1958.10501484.
- Rogers, E.M., 1962. Diffusion of Innovations. Free Press, New York.
- Scheffer, M., Bascompte, J., Brock, W.A., Brovkin, V., Carpenter, S.R., Dakos, V., Held, H., Van Nes, E.H., Rietkerk, M., Sugihara, G., 2009. Early-warning signals for critical transitions. Nature 461, 53–59. https://doi.org/10.1038/nature08227.
- Scheffer, M., Carpenter, S.R., 2003. Catastrophic regime shifts in ecosystems: linking theory to observation. Trends Ecol. Evol. 18, 648–656. https://doi.org/10.1016/ J.TREE.2003.09.002.
- Scheffer, M., Carpenter, S.R., Lenton, T.M., Bascompte, J., Brock, W., Dakos, V., van de Koppel, J., van de Leemput, I.A., Levin, S.A., van Nes, E.H., Pascual, M., Vandermeer, J., 2012. Anticipating Critical Transitions. Science (80-) 338, 344–348. https://doi.org/10.1126/science.1225244.
- Siders, A.R., Hino, M., Mach, K.J., 2019. The case for strategic and managed climate retreat. Science (80-.). 365, 761–763. https://doi.org/10.1126/science.aax8346. Singh, R., Quinn, J.D., Reed, P.M., Keller, K., 2018. Skill (or lack thereof) of data-model fusion techniques to provide an early warning signal for an approaching tipping point. PLoS One 13, 1–21. https://doi.org/10.1371/journal.pone.0191768.
- Smith, E.K., Wiedermann, M., Donges, J.F., Heitzig, J., Winkelmann, R., 2022, under review. Concern and anticipation of sea-level rise provide grounds for social tipping towards climate action.
- Stadelmann-Steffen, I., Eder, C., Harring, N., Spilker, G., Katsanidou, A., 2021. A framework for social tipping in climate change mitigation: What we can learn about social tipping dynamics from the chlorofluorocarbons phase-out. Energy Res. Soc. Sci. 82, 102307 https://doi.org/10.1016/j.erss.2021.102307.
- Steinmann, P., Auping, W.L., Kwakkel, J.H., 2020. Behavior-based scenario discovery using time series clustering. Technol. Forecast. Soc. Change 156, 120052. https://doi.org/10.1016/j.techfore.2020.120052.
- Sterl, A., van den Brink, H., de Vries, H., Haarsma, R., van Meijgaard, E., 2009. An ensemble study of extreme storm surge related water levels in the North Sea in a changing climate. Ocean Sci. 5, 369–378. https://doi.org/10.5194/os-5-369-2009.
- Storlazzi, C.D., Gingerich, S.B., Van Dongeren, A., Cheriton, O.M., Swarzenski, P.W., Quataert, E., Voss, C.I., Field, D.W., Annamalai, H., Piniak, G.A., McCall, R., 2018. Most atolls will be uninhabitable by the mid-21st century because of sea-level rise exacerbating wave-driven flooding. Sci. Adv. 4, 1–10. https://doi.org/ 10.1126/sciadv.aap9741.
- Suzuki, A.H., Zambon, R.C., Yeh, W.-W.-G., 2015. Water supply planning and operation in the Metropolitan Region of Saõ Paulo: Worst drought in history, conflicts, response, and resilience, in. In: World Environmental and Water Resources Congress 2015: Floods, Droughts, and Ecosystems Proceedings of the 2015 World Environmental and Water Resources Congress. https://doi.org/10.1061/9780784479162.219.
- Tàbara, J.D., Frantzeskaki, N., Hölscher, K., Pedde, S., Kok, K., Lamperti, F., Christensen, J.H., Jäger, J., Berry, P., 2018. Positive tipping points in a rapidly warming world. Curr. Opin. Environ. Sustain. 31, 120–129. https://doi.org/10.1016/j.cosust.2018.01.012.
- Tàbara, J.D., Lieu, J., Zaman, R., Ismail, C., Takama, T., 2021. On the discovery and enactment of positive socio ecological tipping points: insights from energy systems interventions in Bangladesh and Indonesia. Sustain. Sci. https://doi.org/10.1007/s11625-021-01050-6.
- Taherkhani, M., Vitousek, S., Barnard, P.L., Frazer, N., Anderson, T.R., Fletcher, C.H., 2020. Sea-level rise exponentially increases coastal flood frequency. Sci. Rep. 10, 6466. https://doi.org/10.1038/s41598-020-62188-4.

Tainter, J.A., 1988. The Collapse of Complex Societies, New Studies in Archaeology. Cambridge University Press, Cambridge.

- Termeer, C.J.A.M., Dewulf, A., Biesbroek, G.R., 2017. Transformational change: governance interventions for climate change adaptation from a continuous change perspective. J. Environ. Plan. Manag. 60, 558–576. https://doi.org/10.1080/09640568.2016.1168288.
- Tesselaar, M., Wouter Botzen, W.J., Haer, T., Hudson, P., Tiggeloven, T., Aerts, J.C.J.H., 2020. Regional inequalities in flood insurance affordability and uptake under climate change. Sustain. 12, 1–30. https://doi.org/10.3390/su12208734.
- Thomas, A., Theokritoff, E., Lesnikowski, A., Reckien, D., Jagannathan, K., Cremades, R., Campbell, D., Joe, E.T., Sitati, A., Singh, C., Segnon, A.C., Pentz, B., Musah-Surugu, J.I., Mullin, C.A., Mach, K.J., Gichuki, L., Galappaththi, E., Chalastani, V.I., Ajibade, I., Ruiz-Diaz, R., Grady, C., Garschagen, M., Ford, J., Bowen, K., 2021. Global evidence of constraints and limits to human adaptation. Reg. Environ. Chang. 21, 85. https://doi.org/10.1007/s10113-021-01808-9.

Tonn, G.L., Guikema, S.D., 2018. An Agent-Based Model of Evolving Community Flood Risk. Risk Anal. 38, 1258–1278. https://doi.org/10.1111/risa.12939.

- Tonn, G., Guikema, S., Zaitchik, B., 2019. Simulating Behavioral Influences on Community Flood Risk under Future Climate Scenarios. Risk Anal. risa.13428 https://doi.org/10.1111/risa.13428.
- Truong, C., Oudre, L., Vayatis, N., 2020. Selective review of offline change point detection methods. Signal Processing 167, 107299. https://doi.org/10.1016/j. sigpro.2019.107299.
- Vaghefi, A.S., Muccione, V., van Ginkel, K.C.H., Haasnoot, M., 2021. Using Decision Making under Deep Uncertainty (DMDU) approaches to support climate change adaptation of Swiss Ski Resorts. Environ. Sci. Policy 126, 65–78. https://doi.org/10.1016/j.envsci.2021.09.005.
- van der Bolt, B., van Nes, E.H., Bathiany, S., Vollebregt, M.E., Scheffer, M., 2018. Climate reddening increases the chance of critical transitions. Nat. Clim. Chang. 8, 478–484. https://doi.org/10.1038/s41558-018-0160-7.
- van Ginkel, K.C.H., Hoekstra, A.Y., Buurman, J., Hogeboom, R.J., 2018. Urban Water Security Dashboard: Systems Approach to Characterizing the Water Security of Cities. J. Water Resour. Plan. Manag. 144, 04018075. https://doi.org/10.1061/(ASCE)WR.1943-5452.0000997.
- van Ginkel, K.C.H., Botzen, W.J.W., Haasnoot, M., Bachner, G., Steininger, K.W., Hinkel, J., Watkiss, P., Boere, E., Jeuken, A., de Murieta, E.S., Bosello, F., 2020. Climate change induced socio-economic tipping points: review and stakeholder consultation for policy relevant research. Environ. Res. Lett. 15, 023001 https:// doi.org/10.1088/1748-9326/ab6395.
- van Ginkel, K.C.H., Koks, E.E., de Groen, F., Nguyen, V.D., Alfieri, L., 2022. Will river floods 'tip' European road networks? A robustness assessment. Transp. Res. Part D Transp. Environ. 108, 10332 https://doi.org/10.1016/j.trd.2022.103332.
- Verburg, P.H., Dearing, J.A., Dyke, J.G., van der Leeuw, S., Seitzinger, S., Steffen, W., Syvitski, J., 2016. Methods and approaches to modelling the Anthropocene. Glob. Environ. Chang. 39, 328–340. https://doi.org/10.1016/j.gloenvcha.2015.08.007.
- Vitousek, S., Barnard, P.L., Fletcher, C.H., Frazer, N., Erikson, L., Storlazzi, C.D., 2017. Doubling of coastal flooding frequency within decades due to sea-level rise. Sci. Rep. 7, 1–9. https://doi.org/10.1038/s41598-017-01362-7.
- Wilby, R.L., Dessai, S., 2010. Robust adaptation to climate change. Weather 65, 180-185. https://doi.org/10.1002/wea.543.
- Winkelmann, R., Donges, J.F., Smith, E.K., Milkoreit, M., Eder, C., Heitzig, J., Katsanidou, A., Wiedermann, M., Wunderling, N., Lenton, T.M., 2022. Social tipping processes towards climate action: A conceptual framework. Ecol. Econ. 192, 107242 https://doi.org/10.1016/j.ecolecon.2021.107242.
- Young, O.R., 2012. Arctic tipping points: Governance in turbulent times. Ambio 41, 75-84. https://doi.org/10.1007/s13280-011-0227-4.